Introduction of Algorithms Information Sheet CS 482 Spring 2002 January 21, 2002

Name: Email: Year:

School and Major:

Campus address and phone:

Previous CS courses at 300-level and above:

I authorize the posting of my grade by Cornell ID number, or by the alternate code that I specify below. I also authorize the return of my homework according to the following system:

Problem sets and prelims will be returned in lecture. Problem sets not picked up in lecture will be left in a common return box in 303 Upson. Prelims not picked up in lecture will be left with Esha Molette in 4119 Upson.

Signature: